Author’s response to reviews

Title: Colposcopy Telemedicine: live versus static Swede score and accuracy in detecting CIN2+, a cross-sectional pilot study

Authors:

Katayoun Taghavi (katayoun.taghavi@ispm.unibe.ch)
Dipanwita Banerjee (dr.dipanwita@yahoo.co.in)
Ranajit Mandal (kranajitmandal@yahoo.co.in)
Helena Kopp Kallner (Helena.Kopp-Kallner@ki.se)
Malin Thorsell (malin.thorsell@bbstockholm.se)
Therese Friis (theresefriis@gmail.com)
Ljiljana Kocoska-Maras (ljiljana.kocoska.maras@akademiska.se)
Björn Strander (bjorn.strander@rccvast.se)
Albert Singer (profasinger@hotmail.com)
Elisabeth Wikström (elisawiksshem@gmail.com)

Version: 2 Date: 09 May 2018

Author’s response to reviews:

Dear Dr Olaitan,

The paper has been revised again and several oversights amended including that which was picked up by you on line 215.

Many thanks for this opportunity

Kind regards

Katayoun Taghavi